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           SOHO Daily Meeting Minutes for Tuesday 24 May 2005

ANNOUNCEMENTS
-------------
* The test data for the SLE test on May 31 will be taken from a subset of
  a recording from 15:49-16:10 UT on 31 August (DOY 244) 2004. The test data
  will be repeated a few times during the test We would still like to have 
  more participants in the test. So far, MDI, UVCS, and CELIAS have signed 
  up.

* The current RHESSI default target is AR10767. When selecting this 
  target to maximize coordination w/others, use campaign number 6850.

* UPS 15 (feeding PDU 15) will undergo corrective maintenance on
  Thursday May 26 starting at 13 UT (9am local). The UPS will be
  in bypass mode during this activity.

FOT REPORT 
-----------
Spacecraft Status: Nominal (APME-A is off)

Spacecraft Anomalies: None

Accomplished Activities: Nominal Daily Activities

Planned Activities: RSL, New Guide Star (6.3 mag)

Upcoming Operations: None

Ground Anomalies: 
143/1409 - Corrupted TC during SVM commanding activities at D27; 
           Cause unknown (No impact) 

SOLAR STATUS & TARGETS
----------------------
See http://soho.nascom.nasa.gov/soc/statustargets.html

INSTRUMENT STATUS
-----------------
CDS:    Nominal. See MEDOC minutes.

SUMER:  Nominal. See MEDOC minutes.

UVCS:   Nominal. JOP158 and synoptics.

LASCO:  Nominal. C2/C3 synoptics.

EIT:    Nominal. Half resolution 195 CME Watch.

MDI:    Full disk magnetograms and Dopplergrams.

TRACE:  Nominal. AR/sunspot study.

http://soho.nascom.nasa.gov/soc/statustargets.html


Planning for Tuesday 24:
-----------------------
SUMER :
01:15 - 13:15 Coronal hole structure - det B (X=0,Y=-1150)
15:00 - 17:00 Detector B changed to detector A
17:00 - 20:00 Coronal hole structure - det A (X=0,Y=-1150)
20:00 - 24:00 TBD

CDS :
00:00 - 07:00 Synoptic
07:00 - 09:00 Monitoring activities
09:00 - 12:00 AR Campaign with VTT & THEMIS (X=-726, Y =-100)
12:00 - 18:00 EUV intensities in AR (Pointing TBD by CDS planner)
18:00 - 24:00 JOP 158 (X=0, Y=-1010)

Planning for Wednesday 25:
-------------------------
SUMER :
May 24, 20:00 - May 25, 06:00 Coronal hole structure - det A (X=0,Y=-1150)
06:00 - 15:00 Filament observations (Brigitte Schmieder) (X=-330,Y=-63)
15:00 - 24:00 Prominence observations (B. Schmieder) (X=970, Y=168)

CDS :
00:00 - 07:00 Synoptic
07:00 - 12:00 AR Campaign with VTT & THEMIS on AR 0767 (X=-562"; Y=-97")
12:00 - 15:00 Filament observations (Brigitte Schmieder)(X=-330,Y=-63)
15:00 - 24:00 Prominence observations (B. Schmieder)(X=970, Y=168)


